Author's response to reviews

Title: Influence of antioxidant (l- ascorbic acid) on tolbutamide induced hypoglycaemia/antihyperglycaemia in normal and diabetic rats

Authors:

Satyanarayana Sreemantula (nandinisai@hotmail.com)
Eswar K Kilari (ekilari@rediffmail.com)
VishnuVardhan A (vishani_pharma@rediffmail.com)
Rajasekhar Jaladi (rajasekhar_27@hotmail.com)

Version: 4 Date: 22 December 2004

Author's response to reviews:


The reference (above) indicates that there is no statistically significant difference in blood glucose or serum insulin levels between male and female diabetic rats.

You can find the same in the attached reference. page no H2689 under the results starting from 6th line.